Author's response to reviews

Title: Lower respiratory tract infection and rapid expansion of an abdominal aortic aneurysm: A case report

Authors:

Steven Naylor (stevennaylor@doctors.org.uk)
Zakareya Gamie (ugm1zg@doctors.org.uk)
Ravinder S Vohra (raysvohra@hotmail.com)
Sapna Puppala (Sapna.Puppala@leedsth.nhs.uk)
Patrick J Kent (Patrick.Kent@leedsth.nhs.uk)
D Julian A Scott (profdjascott@yahoo.co.uk)

Version: 4 Date: 28 May 2010

Author's response to reviews:

Dear editor,

Thank you for comment dated 27/05/10. We now have a conclusion section as the last section of the text. This is as follows:

Conclusion
This case highlights the potential association between pulmonary sepsis and rapid AAA expansion. In these patients there must be a high index of clinical suspicion for rapid progression, and a policy of frequent monitoring may be adopted to identify those patients requiring definitive management. Endovascular repair may be suitable in certain cases depending on aneurysm morphology and local experience.

We have uploaded the updated version of the manuscript.

Kind regards,

Dr Zakareya Gamie